Author's response to reviews

Title: The Role of Advance Directives in End-of-Life Decisions in Austria - Survey of Intensive Care Physicians

Authors:

Eva Schaden (eva.schaden@meduniwien.ac.at)
Petra Herczeg (petra.herczeg@univie.ac.at)
Stefan Hacker (stefan.hacker@meduniwien.ac.at)
Andrea Schopper (andrea.schopper@gmx.at)
Claus Krenn (claus.krenn@meduniwien.ac.at)

Version: 2 Date: 12 April 2010

Author's response to reviews:

Dear Madam,

There were no suggestions how to revise the manuscript so far. Prof Vincent has recommended to target your journal, which suits better to our topic.

Thank you in advance for considering our manuscript for publication.

Yours sincerely,

Eva Schaden